Author’s response to reviews

Title: The efficacy of tumor debulking surgery is improved by adjuvant immunotherapy using imiquimod and anti-CD40

Authors:
Andrea Khong (andrea.khong@uwa.edu.au)
Amanda L Cleaver (amanda.cleaver@uwa.edu.au)
Muhammad Fahmi Alatas (fahmialatasdr@gmail.com)
Ben C Wylie (bwylie@ichr.uwa.edu.au)
Theresa Connor (tconnor@ichr.uwa.edu.au)
Scott A Fisher (Scott.fisher@uwa.edu.au)
Steve Broomfield (S.Broomfield@murdoch.edu.au)
Willem J Lesterhuis (willem.lesterhuis@uwa.edu.au)
Andrew J Currie (A.Currie@murdoch.edu.au)
Richard A Lake (Richard.lake@uwa.edu.au)
Bruce W Robinson (bruce.robinson@uwa.edu.au)

Version: 4
Date: 28 November 2014

Author’s response to reviews: see over
To the Editor,

We provide here a point-by-point response to the Editor’s requests:

- The funding source for the study has been included in the acknowledgements section. All authors are also supported by grants from the National Health and Medical Research Council (NHMRC).

- Figure legends are included for 8 figures (Figures 1-7 plus Supplementary Figure 1). The figures and legends have been checked and correspond with each other.

- Figure 7 was produced entirely by us.

Yours sincerely,

Andrea Khong